Author's response to reviews

Title: Early telemedicine training and counselling after hospitalization in patients with severe chronic obstructive pulmonary disease: A feasibility study.

Authors:

Lisbeth K Rosenbek Minet (lisbeth.minet@rsyd.dk)
Line W Hansen (line.willads.hansen2@rsyd.dk)
Claus D Pedersen (Claus.Duedal.Pedersen@rsyd.dk)
Ingrid Titlestad (ingrid.titlestad@rsyd.dk)
Jette K Christensen (Jette.Kroejgaard@rsyd.dk)
Kristian Kidholm (Kristian.Kidholm@rsyd.dk)
Kathrine Rayce (Kathrine.Rayce@rsyd.dk)
Alison Bowes (a.m.bowes@stir.ac.uk)
Lilian Møllegård (Lilian.Moellegaard@rsyd.dk)

Version: 6 Date: 6 November 2014

Author's response to reviews:

4 November 2014
Author Response
Dear Editor,

We have addressed your comment "I still need the authors to make a small revision to their response on query #5. The use of "five subjects" in their manuscript can be confusing to readers as it was to me. "Subjects" could be interpreted as participants, not as topics/domains that I assume the authors want to say." and changed the word "subjects": "During the focus group interview, participants were asked for their views on five topics that were considered important based on a previous study of the technology …". We hope that the manuscript conforms to the journal style with the changes we have added.

Sincerely,
Lisbeth Minet
Corresponding author

Unit for Rehabilitation at the Institute of Clinical Research, Faculty of Health Sciences, University of Southern Denmark and Department of Rehabilitation, Odense University Hospital, Sdr. Boulevard 29, 5000 Odense C, Denmark
Email: Lisbeth.Minet@rsyd.dk